Author’s response to reviews

Title: Discontinuously supervised aerobic training vs. physical activity promotion in the self-management of type 2 diabetes in older Italian patients: design and methods of the 'TRIPL-A' randomized controlled trial

Authors:
Francesco Lucertini (francesco.lucertini@uniurb.it)
Carlo Ferri Marini (carlo.ferrimarini@uniurb.it)
Davide Sisti (davide.sisti@uniurb.it)
Vilberto Stocchi (vilberto.stocchi@uniurb.it)
Ario Federici (ario.federici@uniurb.it)
Franco Gregorio (franco.gregorio@sanita.marche.it)
Donata Piangerelli (donata.piangerelli@gmail.com)
Carlos Chiatti (carlo.chiatti@gmail.com)
Antonio Cherubini (a.cherubini@inrca.it)
Massimo Boemi (massimo.boemi59@gmail.com)
Fabio Romagnoli (f.romagnoli@inrca.it)
Michela Cucchi (m.cucchi@inrca.it)
Federica d’Angelo (fededang@hotmail.it)
Maria Paola Luconi (m.luconi@inrca.it)
Anna Rita Bonfigli (a.bonfigli@inrca.it)

Version: 1 Date: 18 Dec 2018

Author’s response to reviews:

Dear Prof. Aronin,

Thank you for the editorial comments.
We have removed the table formatting of both the authors' contacts and list of abbreviations.

We have added the contribution of Dr. Boemi.

We have sent the manuscript to a native American English speaking proofreader.

We have changed "trimester" with "three-month period" throughout the manuscript.

We have also uploaded a version of the manuscript where the revisions made are still visible (MS Word track changes tool).

We look forward to hearing from you soon.

Best wishes,

Francesco